Author’s response to reviews

Title: Relationship between psychological health and quality of life of people affected by leprosy in the community in Guangdong province, China: a cross-sectional study

Authors:

Mingzhou Xiong (Lewnexiong@163.com)
Xiaohua Wang (wkh_21773@163.com)
Ting Su (sular_lala@126.com)
Bin Yang (yangbin101@hotmail.com)
Ming Li (13602808178@139.com)
Daocheng Zheng (zhengdaocheng@126.com)

Version: 4 Date: 16 Mar 2019

Author’s response to reviews:

Dear editor

Thanks for your comments about our manuscript. We have stated that "The datasets used and analyzed during the current study available from the corresponding author on reasonable request" in the Availability of data and material part, and changed the contents to "Not Applicable" in Consent for publication part.

Zheng Daocheng